Author’s response to reviews

Title: Abatacept Induced Granulomatous Hepatitis with a Sarcoidosis- like Reaction: A Blinded Trial in Mice

Authors:
Sultan M. Al-Mogairen (salmogairin@ksu.edu.sa)

Version: 2 Date: 18 Apr 2019

Author’s response to reviews:

PHAT-D-18-00122

To : Benjamin Ragen
   Editor
   BMC Pharmacology and Toxicology

From : Dr. Sultan Al Mogairen
   King Saud University, Riyadh
   Kingdom of Saudi Arabia
   salmogairin@ksu.edu.sa
   almogairin@gmail.com

Re : Resubmission of manuscript to BMC Pharmacology and Toxicology

Editor Comments:

I am pleased to inform you that it is potentially acceptable for publication in BMC Pharmacology and Toxicology, once you have carried out some essential revisions: Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Reply:

Dear Editor
I would like to thank you for your positive comments. Considering your respected suggestions, the manuscript was carefully modified and improved.

Full initials have been done in the contributions section, see page 12.

Editor Comments:

In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Reply:

There was no role of the funding body in the design of the study, collection, analysis, and interpretation of data and in writing the manuscript.

Funding section has been changed, see page 11-12.

Thanks Your sincere

Dr S Almogairen